Author’s response to reviews

Title: The second patient? Family members of cancer patients and their role in end-of-life decision making

Authors:

Katsiaryna Laryionava (Katsiaryna.Laryionava@med.uni-heidelberg.de)
Timo Pfeil (timopfeil@hotmail.com)
Mareike Dietrich (mareike.dietrich@med.uni-heidelberg.de)
Stella Reiter-Theil (stella.reiter-theil@usb.ch)
Wolfgang Hiddemann (wolfgang.hiddemann@med.uni-muenchen.de)
Eva Winkler (Eva.Winkler@med.uni-heidelberge.de)

Version: 3 Date: 08 Feb 2018

Author’s response to reviews:

We are very glad about positive Feedback of the Reviewer and the Editor. Below you will find the comment in normal font and our answers in italic font.

Comments to Author:

1. Comment: 1. If abbreviations are used in the text they should be defined in the text at first use, and a list of abbreviations should be provided in the Declarations.

1. Answer: We added “Abbreviations” in the Declaration part. P.22.

2. Comment: Please represent authors’ names using their full initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.

2. Answer: In the Authors’ Contribution section the full initials of the authors have been added. P. 22-23.
3. Comment: At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

3. Answer: All tracked changes have been accepted. The paper has been formatted accordingly.